Author's response to reviews

Title: Measuring the health impact of human rights violations related to Australian asylum policies and practices: a mixed methods study

Authors:

Vanessa Johnston (vanessa.johnston@menzies.edu.au)
Pascale Allotey (pascale.allotey@brunel.ac.uk)
Kim Mulholland (Kim.Mulholland@lshtm.ac.uk)
Milica Markovic (milica.markovic@med.monash.edu.au)

Version: 2 Date: 3 December 2008

Author's response to reviews: see over
Editor
*BMC International Health and Human Rights*

3rd December, 2008

Dear Editor,

Please find attached the revised manuscript ‘*Measuring the health impact of human rights violations related to Australian asylum policies: a mixed methods study*’ for consideration.

The revisions made to this paper are presented in tracked changes for ease of recognition. We also attach a document detailing our responses to the reviewers of this paper.

Yours sincerely,

Dr. Milica Markovic

Submitting author

Senior Research Fellow
School of Psychology, Psychiatry and Psychological Medicine
Monash University

Phone: 61 3 9903 4049
Email: milica.markovic@med.monash.edu.au